\documentstyle[12pt,fleqn]{article}
\pagestyle{empty}
\setlength{\topmargin}{-1in}
\setlength{\oddsidemargin}{-.3in}
\setlength{\textheight}{9in}
\setlength{\textwidth}{7.2in}

\begin{document}


Math 105 - 4

Quiz 4
\vspace{.1in}
\begin{enumerate}

\item  Name \rule{5.in}{.3pt}

\item  (10 points)  Solve the system of equations
\[
\begin{array}{rrrrrrr}
x & + & y & + & z & = & 2 \\
-x & + & 3y & + & 2z & = & 8 \\
4x & + & y &  &  & = & 4 \\
\end{array}
\]
\vspace{3.5in}

\item  (10 points)  Find the partial fraction decomposition for the following:
\[
\frac{5}{x^2+x-6}
\]







\end{enumerate}


\end{document}






